Author's response to reviews

Title: Family history of diabetes and clinical characteristics in Greek subjects with type 2 diabetes

Authors:

Athanasia Papazafiropoulou (pathan@ath.forthnet.gr)
Alexios Sotiropoulos (alesot@ath.forthnet.gr)
Eystathios Skliros (eskliros@otenet.gr)
Marina Kardara (mkardara@yahoo.gr)
Anthi Kokolaki (anthi_xios@yahoo.gr)
Ourania Apostolou (s.pappas@nikaia_hosp.gr)
Stavros Pappas (el.pappa@yahoo.gr)

Version: 5 Date: 9 April 2009

Author's response to reviews:

Dear Editor,

I have made the revisions according to your requests.

1. I changed the format of the tables.
2. I added the country to the affiliation.
3. I changed the emails of two co-authors (O.Apostolou and S.Pappas)
4. I deleted the keywords from the abstract section.

Kindly yours,
A.Papazafiropoulou